Author's response to reviews

**Title:** Saikosaponin-d Increases the Radiosensitivity of SMMC-7721 Hepatocellular Carcinoma Cells by Adjusting the G0/G1 and G2/M Checkpoints of the Cell Cycle

**Authors:**

wang baofeng (wangbf1680@126.com)
Zhi-Jun Dai (dji0911@126.com)
Xi-Jing Wang (wangxijing@21cn.com)
Ming-Hua Bai (baimh2004@126.com)
Shuai Lin (linshuai420@stu.xjtu.edu.cn)
Hong-Bing Ma (mahongbingxa@126.com)
Ya-Li Wang (wangyili@mail.xjtu.edu.cn)
Ling-Qin Song (songlingqin@163.com)
Xiu-Long Ma (583513169@qq.com)
Ying Zan (zzqq76@sina.com)
Wei-Li Min (mwli791218@126.com)
Yan-An Cheng (cheng.yanan@163.com)

**Version:** 4 **Date:** 2 October 2013

**Author's response to reviews:** see over
Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “Saikosaponin-d Increases the Radiosensitivity of SMMC-7721 Hepatocellular Carcinoma Cells by Adjusting the G0/G1 and G2/M Checkpoints of the Cell Cycle” (MS: 8842196221023094). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have revised the manuscript, and would like to re-submit it for your consideration. Point by point responses to the reviewers’ comments are listed below this letter.

We would like to express our great appreciation to you and reviewers for comments on our paper.

Looking forward to hearing from you.

With best wishes,

Yours sincerely,

Baofeng Wang
2013.10.02

1. **Response to comments: For figure 4, please make a better alignment among each blot.**

Response: Thank you for your question. We have made the necessary corrections in revised paper according to the reviewer’s comments.

We have revised and improved the contents of this manuscript. These changes will not influence the content and framework of the paper. Thank you very much for your comments and suggestions.